

Improved Representation of Clouds in the Atmospheric Component LMDZ6A of the IPSL-CM6A Earth System Model

Jean-baptiste Madeleine, Frédéric Hourdin, Jean-yves Grandpeix, Catherine Rio, Jean-Louis Dufresne, Etienne Vignon, Olivier Boucher, Dimitra Konsta, Frederique Cheruy, Ionela Musat, et al.

▶ To cite this version:

Jean-baptiste Madeleine, Frédéric Hourdin, Jean-yves Grandpeix, Catherine Rio, Jean-Louis Dufresne, et al.. Improved Representation of Clouds in the Atmospheric Component LMDZ6A of the IPSL-CM6A Earth System Model. Journal of Advances in Modeling Earth Systems, 2020, 12 (10), 10.1029/2020MS002046. hal-03084242

HAL Id: hal-03084242 https://hal.science/hal-03084242

Submitted on 20 Dec 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

ACCU ADVANCI EARTH AN SPACE SCIENCE

JAMES Journal of Advances in Modeling Earth Systems

RESEARCH ARTICLE

10.1029/2020MS002046

Special Section:

The IPSL climate model used in CMIP6**

Key Points:

- Cloud parameterizations of the LMDZ6A climate model are entirely described
- Low-level and mid-level cloud distribution and radiative effects are improved compared to LMDZ5A
- LMDZ6A is better tuned than LMDZ5A, and knowledge of its structural deficiencies has been gained

Correspondence to:

J.-B. Madeleine,

jean-baptiste.madeleine@sorbonneuniversite.fr

Citation:

Madeleine, J.-B., Hourdin, F., Grandpeix, J.-Y., Rio, C., Dufresne, J.-L., Vignon, E., et al. (2020). Improved representation of clouds in the atmospheric component LMDZ6A of the IPSL-CM6A Earth system model. *Journal of Advances in Modeling Earth Systems*, *12*, e2020MS002046. https:// doi.org/10.1029/2020MS002046

Received 5 JAN 2020 Accepted 5 JUL 2020 Accepted article online 26 AUG 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Improved Representation of Clouds in the Atmospheric Component LMDZ6A of the IPSL-CM6A Earth System Model

Jean-Baptiste Madeleine¹, Frédéric Hourdin¹, Jean-Yves Grandpeix¹, Catherine Rio², Jean-Louis Dufresne¹, Etienne Vignon³, Olivier Boucher⁴, Dimitra Konsta⁵, Frédérique Cheruy¹, Ionela Musat¹, Abderrahmane Idelkadi¹, Laurent Fairhead¹, Ehouarn Millour¹, Marie-Pierre Lefebvre², Lidia Mellul¹, Nicolas Rochetin¹, Florentin Lemonnier¹, Ludovic Touzé-Peiffer¹, and Marine Bonazzola¹

¹Sorbonne Université, Laboratoire de Météorologie Dynamique/Institut Pierre-Simon Laplace, Centre National de la Recherche Scientifique, École Polytechnique, École Normale Supérieure, PSL Université, Institut Polytechnique de Paris, Campus Pierre et Marie Curie, Paris, France, ²Centre national des recherches météorologiques, Université de Toulouse, Météo-France, CNRS, Toulouse, France, ³Environmental Remote Sensing Laboratory (LTE), École Polytechnique Fédérale de Lausanne, Lausanne, Switzerland, ⁴Institut Pierre-Simon Laplace, Sorbonne Université/CNRS, Paris, France, ⁵Laboratory of Climatology and Atmospheric Environment, National and Kapodistrian University of Athens, Athens, Greece

Abstract The cloud parameterizations of the LMDZ6A climate model (the atmospheric component of the IPSL-CM6 Earth system model) are entirely described, and the global cloud distribution and cloud radiative effects are evaluated against the CALIPSO-CloudSat and CERES observations. The cloud parameterizations in recent versions of LMDZ favor an object-oriented approach for convection, with two distinct parameterizations for shallow and deep convection and a coupling between convection and cloud description through the specification of the subgrid-scale distribution of water. Compared to the previous version of the model (LMDZ5A), LMDZ6A better represents the low-level cloud distribution in the tropical belt, and low-level cloud reflectance and cover are closer to the PARASOL and CALIPSO-GOCCP observations. Mid-level clouds, which were mostly missing in LMDZ5A, are now better represented globally. The distribution of cloud liquid and ice in mixed-phase clouds is also in better agreement with the observations. Among identified deficiencies, low-level cloud covers are too high in mid-latitude to high-latitude regions, and high-level cloud covers are biased low globally. However, the cloud global distribution is significantly improved, and progress has been made in the tuning of the model, resulting in a radiative balance in close agreement with the CERES observations. Improved tuning also revealed structural biases in LMDZ6A, which are currently being addressed through a series of new physical and radiative parameterizations for the next version of LMDZ.

Plain Language Summary This paper describes the representation of clouds in the latest version of LMDZ, which is a French atmospheric model used for climate change projections. Along with other international climate models, it serves as a basis for the IPCC (Intergovernmental Panel on Climate Change) report by contributing to the CMIP project (Climate Model Intercomparison Project). Clouds are especially important in the climate system because they reflect a lot of sunlight and also absorb and emit a lot of infrared radiation. They can either amplify or reduce the current global warming depending on their change in opacity, altitude, and detailed properties. It is therefore essential to represent them accurately in climate models. The main physical equations used to compute cloud properties in LMDZ are introduced, and the model results are compared to various satellite observations. It reveals that low-level and mid-level clouds are in better agreement with the observations than before but that high-level clouds remain difficult to simulate realistically. Ongoing developments aimed at solving these remaining deficiencies are finally described.

1. Introduction

On average, two thirds of the Earth's surface is covered by clouds (King et al., 2013), which are therefore of primary importance in the energy budget of the atmosphere. Similarly, cloud response to global



warming is one of the largest sources of uncertainty in climate change simulations (Bony et al., 2006; Dufresne & Bony, 2008; Vial et al., 2013). From the early stages of climate modeling at the "Laboratoire de Météorologie Dynamique" (LMD) (Sadourny, 1975; Sadourny & Laval, 1984; Laval et al., 1981), efforts were made to develop innovative subgrid-scale parameterizations that correctly represent their effect (Le Treut & Li, 1991; Li, 1999). The current LMD global atmospheric model, called LMDZ for its zooming capability (Hourdin et al., 2006), is the atmospheric component of the Earth system model named after the French climate institute where it is developed: the Institut Pierre-Simon Laplace (IPSL) Climate Model or IPSL-CM. This paper describes the representation of clouds in the latest version of LMDZ, LMDZ6A, which was used for the sixth phase of the Coupled Model Intercomparison project (CMIP6, Eyring et al., 2015). The general descriptions of the IPSL-CM6A model and its atmospheric component, LMDZ6A, can be found in this Special Collection, in two papers by Boucher et al. (2020) and Hourdin et al. (2020), respectively. The two previous versions of LMDZ were LMDZ5A and 5B and are described in Hourdin, Foujols, et al. (2013) and Hourdin, Grandpeix, et al. (2013). Compared to Version 5A, Version 5B was based on a profound rethinking of the parameterization of convection and clouds, on which the new 6A version is built. In the present paper, we will focus on comparing LMDZ5A with LMDZ6A directly, because Version 5B was in many respects the prototype of Version 6A. CMIP5 revealed a number of biases in LMDZ5A cloud properties:

- Despite major development efforts, tropical and subtropical low-level cloud fractions were underestimated;
- · Mid-level clouds were almost inexistent;
- Large biases were found in the total cloud radiative effect (CRE) over the Southern Ocean;
- Low-level cloud cover was underestimated, and cloud reflectance was overestimated (Konsta et al., 2016);
- The altitude of low-level clouds was too low (Konsta et al., 2016).

Our goal in this paper is twofold: to review the entire set of cloud parameterizations developed for LMDZ, and to present the main improvements of the newest version, LMDZ6A. Particular care has been given in the LMDZ parameterizations to the representation of convection, for which a deliberate choice was made to separate deep and shallow convection and which is coupled to cloud description through the specification of subgrid-scale distribution of total water or saturation deficit. These developments have been described through a series of publications but always focusing on one particular aspect. The present paper provides a full description of the parameterizations that control clouds in LMDZ as well as their interactions. In terms of evaluation, particular attention will be paid to the global cloud distribution and its role in maintaining the global radiative balance in the model. The discussion and conclusion will highlight the remaining biases and present the current development efforts to address them.

2. Parameterization of Clouds in LMDZ6A

The challenge in modeling clouds resides in the various scales of atmospheric processes controlling their macrophysical and microphysical properties. They depend on both km-scale and μ m-scale processes evolving on time scales ranging from minutes to seconds. In the last two decades, the LMDZ team worked on a set of innovative parameterizations that describe the subgrid-scale vertical motions and their connections to cloud properties. Clouds in LMDZ depend on (1) turbulent mixing, shallow convection, deep convection, and large-scale horizontal advection, and (2) cloud statistical schemes that use the physical information provided by these processes to compute their opacity and the fraction of the gridbox they cover. To do so, atmospheric properties such as the area covered by thermal plumes in the boundary layer or mass fluxes in deep convective clouds are used to shape the subgrid-scale distributions of water vapor. The general approach is to represent these distributions by probability density functions (PDFs) that can be unimodal or bimodal and whose variance and asymmetry toward high humidity values increases when convective plumes bring near-surface moist air toward the drier free troposphere. Since the temperature of the gridbox is known, it is possible to derive, from these distributions, the populations of air parcels that are supersaturated and to deduce the cloud fraction and water content.

All the processes occurring in a gridbox (turbulent mixing, shallow and deep convection) are called sequentially in LMDZ, as represented in Table 1. In the following sections, the different steps of this diagram will be described, from the main model prognostic variables to the final cloud fraction α_c and water



Architecture of the Physical Package, Showing All Cloud-Related Variables						
Procedure / Subsection	Input variables	Other outputs				
	() Updated variables					
2.1. Evaporation	$\theta q_v q_l q_i$ (5) $\theta q_t (q_l = q_i = 0)$					
2.2. Local turbulent mixing	$\begin{array}{c} \bigcirc \ \lor \ q_t \ (q_t = q_t = 0) \end{array}$					
2.3. Deep convection		$q_{a}^{in,cv} P_{La}^{cv} d\theta_{du}^{cv} dq_{t,du}^{cv}$				
	$\bigcirc \theta q_t$					
2.4. Deep convection PDF	$q_t \; q_c^{in,cv}$	$lpha_c^{cv}$				
2.5. Cold pools (wakes)	$\theta q_t d\theta_{dw}^{cv} dq_{t,dw}^{cv}$	$ALE^{wk} \ ALP^{wk} \ \theta^{wk}_{env} \ q^{wk}_{t,env}$				
2.6. Shallow convection	$\theta_{env}^{wk} q_{t,env}^{wk}$	$(s_{th} \sigma_{th} s_{env} \sigma_{env})^{th} ALE^{th} ALP^{th}$				
2.7. Large-scale condensation	$\theta q_t (s_{th} \sigma_{th} s_{env} \sigma_{env})^{th}$	$q_c^{in,lsc} \; lpha_c^{lsc} \; P_{l,i}^{lsc}$				
	\circlearrowleft θ q_v q_l q_i					
2.8. Radiative transfer	$q_c^{in,lsc} \; \alpha_c^{lsc} \; q_c^{in,cv} \; \alpha_c^{cv}$					
	\circ θ					

Table 1

Note. The first column gives the names of the different procedures, that are also used as subsection titles in section 2. The second column indicates the main variables used by the procedure on the left, and the prognostic variables that are updated at the end of the procedure on the right, in gray. The other useful variables computed by each procedure are given in the last column. Variables colored in blue are related to cloud properties, and are those used by the radiative transfer scheme to compute the cloud radiative effect. All the notations are given in the text and summarized in Appendix A.

content q_{c}^{in} , which are the two information used by the radiative transfer scheme to compute cloud radiative heating rates.

2.1. Evaporation

The first procedure of the LMDZ physical package is the evaporation of all condensates, because most parameterizations of convection work with the total water mass mixing ratio q_t (see the early work of Betts, 1973). This does not mean that clouds are purely diagnostic. The cloud liquid and ice mixing ratios are "semi-prognostic" variables in the sense that they are advected by the dynamical core, but they are evaporated/sublimated at each time step at the beginning of the physical package. This assumption may hold for liquid droplets whose lifetime is often smaller than the physics time step of ~15 min but can be a limitation for ice or mixed-phase clouds. This first procedure is represented in Table 1 and affects the three water phases (the water vapor, liquid water, and ice mass mixing ratios, noted q_v , q_l , and q_i) as well as the potential temperature θ , through evaporative cooling. It returns the total water content q_t which is then used and updated by all the cloud parameterizations. The only other procedure affecting the prognostic variables q_i and q_i is the so-called large-scale condensation scheme, which condenses, before calling the radiative transfer scheme, all the water vapor in excess of saturation coming from the different parameterizations.

2.2. Local Turbulent Mixing

The first process that is accounted for is the local turbulent mixing in the boundary layer, which was revisited in LMDZ6A. It now includes a 1.5 order closure K-gradient scheme and a prognostic equation for the TKE (turbulent kinetic energy). The K-gradient scheme is based on the work of Yamada (1983) and was improved for stable boundary layers (Cheruy et al., 2020; Vignon et al., 2017). The total water vapor mass mixing ratio q_t is vertically mixed assuming a down-gradient Fick's type diffusion whose intensity



depends on the TKE. As is classical in climate models, the turbulence scheme includes the representation of exchanges with the surface, including the evaporation and sensible heat fluxes, which are essential to cloud formation.

2.3. Deep Convection

The deep convection scheme of LMDZ computes heating, moistening, and momentum changes using a modified version of the Emanuel mass flux scheme (Emanuel, 1991) to which a parameterization of cold pools was added (Grandpeix & Lafore, 2010; Grandpeix, Lafore & Cheruy 2010). Version 6A differs significantly from Version 5A which was using the Emanuel scheme without the improved mixing representation (Grandpeix et al., 2004) and the various improvements described in the present section.

Once the turbulent mixing in the boundary layer has been computed, deep convection can be initiated, depending on the *ALE* (available lifting energy) inherited from the previous time step. The *ALE* can be provided by frontal lifting at the edge of cold pools or by boundary layer thermals, which are noted ALE^{wk} and ALE^{th} in Table 1, respectively. The *ALE* finally used by the deep convection scheme is the largest of the two energies. Deep convection is triggered if the *ALE* exceeds the *CIN* (Convective INhibition) and if at least one of the cumulus of the domain reaches a given threshold size and evolves into a congestus or cumulonimbus cloud. This latter process is represented by a stochastic triggering scheme (Rochetin et al., 2014) and is also a new feature of LMDZ6A. Another important new feature of Version 6A is the inclusion of the latent heat exchange due to the liquid \leftrightarrow ice phase change in the deep convection scheme.

Once deep convection has been triggered and the cold pools have been initiated, the column is split into two separate fields: the cold pool area and its environment, each having their own temperature and humidity. Deep convection then "sees" the environment of cold pools, rather than the mean grid cell, while downdrafts fall inside the cold pool region. This so-called splitting technique is essential to maintain deep convection within the grid cell. The deep convection closure is based on the *ALP* (available lifting power; see Grandpeix, Lafore & Cheruy 2010), which is inherited from the previous time step and is the sum of the *ALP* provided by the cold pools and by the thermal plumes of the boundary layer.

The deep convection scheme then computes the in-cloud water mass mixing ratio $q_c^{in, cv}$, which is the ratio of condensed water mass to *in-cloud* air mass. Note that this quantity is different from the liquid or ice mass mixing ratios within a gridbox q_l and q_i which correspond to the ratio of condensed water mass to *gridbox* air mass. It also computes the convective rainfall and snowfall $P_{l,i}^{cv}$. The precipitation mechanism follows Emanuel and Ivkovi-Rothman (1999): All the condensate in excess of a temperature-dependent conversion threshold is converted into large hydrometeors that will eventually fall. The precipitation efficiency (i.e., the fraction of large hydrometeors in the total condensate) is bounded by a maximum value ep_{max} , which is usually slightly lower than 1 (see Table 3) to always keep some cloud water in the atmosphere (Bony & Emanuel, 2001). All the condensate is carried up in the updrafts till their ends, at which point the large hydrometeors fall as precipitation with a prescribed terminal velocity.

In our scheme, both the undiluted updrafts and the mixed drafts contribute to the in-cloud water content of deep convective clouds. The deep convective cloud fraction α_c^{cv} is computed (as explained in section 2.4) from the in-cloud water content of deep convective clouds $q_c^{in,cv}$, which is itself deduced from the different mass fluxes and coverage fraction of undiluted and mixed updrafts. In the case of undiluted updrafts, the coverage fraction α_a is given by $\alpha_a = M_a/(\rho w_a)$ where M_a is the mass flux density and w_a the vertical velocity. In the case of the mixed drafts, the entrained air at each level feeds cloud formation, and these clouds dissipate with a time constant τ_m . Therefore, the time evolution of the cloud water mass in a layer of thickness δz can be written as

$$\frac{\partial}{\partial t}(\rho \alpha_m \delta z q_m) = M_t q_m - \frac{\rho \alpha_m \delta z q_m}{\tau_m},\tag{1}$$

where M_t is the mass flux density of the mixed drafts and q_m its condensed water mixing ratio. The coverage fraction of mixed drafts can then be deduced from Equation 1 by assuming a steady state, which gives $\alpha_m = M_t \tau_m / (\rho \delta z)$. The in-cloud water content is finally calculated as a linear combination of the cloud water of the undiluted updraft and mixed drafts:



$$q_c^{in,\,cv} = \frac{\alpha_a q_a + \alpha_m q_m}{\alpha_a + \alpha_m},\tag{2}$$

where q_a is the condensed water mixing ratio of the undiluted updraft. In Equation 1, the saturated draft dissipates with a time constant τ_m of the order of 100 s.

This in-cloud water content $q_c^{in,cv}$ is computed for use in the radiative transfer scheme and in the deep convective cloud statistical scheme (see the next section), but it is not removed from the vapor phase or used to derive the prognostic variables q_l and q_i . At the end of the deep convection scheme, the vertical profiles of convective rainfall and snowfall $P_{l,i}^{cv}$ are returned and removed from the vapor phase, and only θ and the total water mass mixing ratio q_t are changed accordingly. The deep convection scheme also returns the change in both temperature and water content due to downdrafts $d\theta_{dw}^{cv}$ and $dq_{t,dw}^{cv}$, which are later used by the cold pool scheme (see section 2.5).

2.4. Deep Convection PDF

As briefly mentioned at the beginning of this section, the cloud statistical schemes used in LMDZ are tightly connected to the information provided by the shallow and deep convection schemes. Such statistical schemes rely on a PDF describing the subgrid-scale distribution of water vapor or saturation deficit. In the case of deep convection, the total mass mixing ratio of water q within the gridbox is assumed to be a random variable of mean value q_t . The latter can be written as

$$q_t = \int_0^\infty q P(q) \, dq. \tag{3}$$

The cloud mixing ratio q_l or q_i and cloud fraction α_c can then be computed as

$$q_{l,i} = \int_{q_{sat}}^{\infty} (q - q_{sat}) P(q) dq, \text{ and}$$
(4)

$$\alpha_c = \int_{q_{sat}}^{\infty} P(q) \, dq, \tag{5}$$

where q_{sat} is the water vapor saturation mixing ratio at the gridbox mean temperature and pressure, that is, $q_{sat}(T, p)$. We neglect in this case the effect of temperature heterogeneities on q_{sat} . The gridbox mean amount of both condensates and in-cloud vapor, q_{t_r} , can be written as

$$q_{t_c} = \int_{q_{sat}}^{\infty} q P(q) \, dq, \text{ with}$$
(6)

$$q_{t_c} = q_{l,i} + \alpha_c q_{sat}. \tag{7}$$

In this context, the in-cloud water content q_c^{in} is given by

$$q_c^{in} = \frac{\int_{q_{sat}}^{\infty} (q - q_{sat}) P(q) dq}{\int_{q_{out}}^{\infty} P(q) dq} = \frac{q_{l,i}}{\alpha_c}.$$
(8)

The deep convection scheme provides the in-cloud water content $q_c^{in, cv}$, as described in section 2.3. Therefore, the three free parameters of a lognormal PDF are then deduced from Equations 3 and 8 by an inverse procedure, assuming that the PDF equals 0 for q = 0 (Bony & Emanuel, 2001, Appendix A). The PDF is then used to compute α_c^{cv} , which is later used, together with $q_c^{in, cv}$, by the radiative transfer scheme (see Table 1).

2.5. Cold Pools (Wakes)

Density currents are outflows of evaporatively cooled downdrafts generated in thunderstorms and larger convective systems. They result in surface cold pools that inhibit convection locally on the one hand, but



favor new convective zones at their edges on the other hand. Therefore, they play an important role in the lifecycle of convective systems. Their representation is a new feature of LMDZ6A. To account for this process, the deep convection scheme assumes that a fraction of precipitation (15% above cloud base and 100% below) falls outside the cloud and evaporate to form precipitating downdrafts. The cold pool scheme then uses the change in both temperature and water content due to these downdrafts $d\theta_{dw}^{cv}$ and $dq_{t,dw}^{cv}$. As explained earlier, we use a splitting technique so that cold pools can have their own temperature and humidity. The cold pool scheme also derives its own *ALE* and *ALP* quantities that will be later used, at the next time step, by the deep convection scheme for its triggering and closure (Grandpeix, Lafore & Cheruy 2010). Density currents affect clouds indirectly in two ways. First, they redistribute heat and water vapor vertically. Second, they play a role, via the term ALE^{wk} , in triggering deep convection.

2.6. Shallow Convection

2.6.1. Thermal Plume Model and Shallow Cumulus Convection

Version 6A uses a mass flux parameterization of thermals (Hourdin et al., 2002) instead of using a countergradient term in the vertical derivative of potential temperature and a dry convective adjustment as was the case in Version 5A (Hourdin, Foujols, et al., 2013). This thermal plume model was extended to the representation of shallow cumulus convection by Rio and Hourdin (2008). Conceptually, this model represents two subgrid-scale objects: a given coverage fraction of thermals, and their environment. The splitting technique mentioned in the previous section is also applied to the shallow convection scheme and thermals develop outside the cold pool region and in the same environment as the convective updrafts, that is, in a more unstable environment than that of the mean atmospheric grid cell. To do so, the potential temperature and total water content outside the cold pool region (θ_{env}^{wk} and $q_{t,env}^{wk}$ in Table 1) are used as inputs of the shallow convection scheme, thereby improving the buoyancy calculations and thermals development. In LMDZ6A, the thermal plume model was also improved by changing the detrainment formulation to better represent the transition from stratocumulus to cumulus clouds. This was done by using in the buoyancy formulation the difference in virtual potential temperature between the updraft and the environment at two different vertical levels, instead of computing the temperature difference on a same level. This method significantly improved the representation of clouds in regions of subsidence (for more details, see Hourdin et al., 2019).

2.6.2. Statistical Cloud Scheme

The shallow convection scheme is tightly connected to a statistical cloud scheme that uses a bi-Gaussian distribution Q of the saturation deficit s (Jam et al., 2013). The parameters required to compute the bi-Gaussian distribution are given by the thermal plume scheme and provided to the so-called large-scale condensation scheme described in the next section. In order to partly account for subgrid-scale temperature fluctuations, each Gaussian distribution is characterized by the mean saturation deficit and standard deviation of the thermal plume (s_{th} and σ_{th}) and its environment (s_{env} and σ_{env}), where the environment corresponds to the main mode of the bimodal distribution. The bi-Gaussian PDF can therefore be written as

$$Q(s) = (1 - \alpha_{th})f(s, s_{env}, \sigma_{env}) + \alpha_{th}f(s, s_{th}, \sigma_{th}),$$
(9)

where α_{th} is the coverage fraction of thermals and *f* is the classical Gaussian PDF:

$$f(s, \bar{s}, \sigma_s) = \frac{1}{\sigma_s \sqrt{2\pi}} \exp\left(\frac{-(s-\bar{s})^2}{2\sigma_s^2}\right).$$
(10)

The in-cloud water content and cloud fraction can then be expressed as

$$q_c^{in} = \int_0^\infty s Q(s) \, ds, \text{ and } \alpha_c = \int_0^\infty Q(s) \, ds.$$
(11)

The two mean saturation deficits s_{th} and s_{env} are computed automatically by the thermal plume model, and the variances are parameterized based on the coverage fraction of thermals α_{th} (see Equations 7 and 8 of Jam et al., 2013). The shallow convection scheme does not remove the condensates from the prognostic total water variable at this stage, and only contributes to the mixing of q_t (see Table 1). Shallow convective





Figure 1. Diagram illustrating the two ways of computing α_c^{ev} in the rain/ snow reevaporation scheme (see section 2.7). Blue and red bars show the actual cloud fraction α_c and precipitation flux density $P_{l,i}$ simulated by the model, respectively. The dashed and solid lines show the cloud fraction α_c^{ev} used to compute the maximum amount of reevaporated rain/ snow in LMDZ5A and LMDZ6A.

cloud formation and conversion to precipitation is computed afterward by the large-scale condensation scheme.

2.7. Large-Scale Condensation

The role of the large-scale condensation scheme is to condense the water vapor in excess of saturation coming from all the other procedures, as well as the water vapor brought to saturation by the large-scale horizontal circulation (which obviously affects q_t and θ as well). It is in charge of the final calculation of the prognostic variables q_l and q_i and rebuilds the cloud macrophysical properties q_c^{in} and α_c for further use in radiative transfer computations. It also computes the large-scale rainfall and snow-fall rates P_l^{lsc} and P_i^{lsc} . Note that the term "large-scale" is a bit abusive in the sense that the cloud amounts and rainfall/snowfall rates computed by the large-scale condensation scheme include both large-scale clouds and shallow cumulus and stratocumulus clouds associated with the thermal plume model.

In practice, the large-scale condensation scheme computes, for each atmospheric column, the different processes using a vertical top-to-bottom loop. In this section, the current layer will be referred to as z_k , with

 z_{k+1} the overlying layer and z_{k-1} the underlying layer. The procedure computes all the condensed water contents in three steps: (1) It computes the reevaporation/sublimation of rain/snow coming from the overlying level z_{k+1} (simply called reevaporation hereinafter), (2) it computes the amount of clouds that form in the gridbox at level z_k using a subgrid-scale PDF, and (3) it converts part of the cloud into rain/snow. These three tasks are performed sequentially in this order. No structural changes were made to this scheme between Versions 5A and 6A, but many existing parameterizations were improved, and these adjustments will be noticed in the following subsections.

2.7.1. Step 1: Reevaporation

The loop starts with the reevaporation at level z_k of the rain or snow coming from level z_{k+1} . This reevaporation is based on the work by Klemp and Wilhelmson (1978) and Schlesinger et al. (1988) and can be written as

$$\frac{\partial P_{l,i}}{\partial z} = \beta \left(1 - \frac{q_t}{q_{sat}} \right) \sqrt{P_{l,i}},\tag{12}$$

where $P_{l,i}$ is the liquid or solid precipitation mass flux density in kg m⁻² s⁻¹. It depends on the relative humidity q_t/q_{sat} and on a parameter called β , which is the same for rain and snow in LMDZ. Reevaporation is such that water vapor in the fraction of the gridbox below clouds does not exceed the saturation mixing ratio. In LMDZ5A, the reevaporation at level z_k is limited to $\alpha_c^{ev}(q_{sat} - q_t)$, where $\alpha_c^{ev}(z_k)$ $= \alpha_c(z_{k+1})$, with α_c the actual cloud fraction simulated by the model (see the dashed line in Figure 1). This means that at two levels below cloud base, α_c^{ev} is set to 0 and reevaporation is no more possible. In LMDZ6A, α_c^{ev} was changed to the maximum cloud fraction found in the overlying layers and is reset back to 0 only if precipitation at level z_{k+1} stops (see the solid line in Figure 1). This method implies that reevaporation is more efficient in Version 6A than in Version 5A (see the shaded gray area in Figure 1), if of course the value of the β coefficient in Equation 12 is unchanged.

2.7.2. Step 2: Cloud Formation

Cloud formation comes next, and the computation of the amount of condensates differs whether shallow convection is active in the gridbox or not. If shallow convection is active, cloud amount and fraction are computed using the bi-Gaussian PDF described in section 2.6. To do so, it uses the mean saturation deficits (s_{th} , s_{env}) and standard deviations (σ_{th} , σ_{env}) computed by the shallow convection scheme (Table 1). Otherwise, outside the grid cells where shallow convection is active, $q_c^{in, lsc}$ and α_c^{lsc} are computed using a generalized lognormal PDF whose standard deviation σ is computed as $\sigma = \xi q_t$. ξ is a function of pressure that has changed through the different versions of the model, as shown in Figure 2. In all versions, ξ is chosen so as to increase from the bottom of the troposphere to the top. Indeed, in the low and middle troposphere, the





Figure 2. $\xi(p)$ profiles used in the two versions of LMDZ. ξ is used to impose the standard deviation σ of the large-scale cloud PDF, with $\sigma = \xi q_t$. The asymptotic value in the upper troposphere, noted ξ_{300} , is a tuning parameter.

shallow convection scheme already computes the subgrid-scale water distributions and the large-scale standard deviation σ is therefore kept close to 0. In the case where the shallow convection scheme is not active, the standard deviation σ being close to 0, the scheme is almost equivalent to an "all-or-nothing" cloud scheme. The variance of the lognormal PDF in the lower and middle troposphere was set to a higher value in LMDZ5A than in LMDZ6A (using the ξ parameter represented in Figure 2) because the bi-Gaussian PDF was not implemented at the time and shallow convective clouds had to be represented by the lognormal PDF. In LMDZ6A, this becomes useless and the variance of the lognormal PDF is strongly reduced in the lower and middle troposphere to let the bi-Gaussian PDF of the shallow convection scheme do the calculation. In the high troposphere, ξ increases to reach a maximum value ξ_{300} , which is used as a tuning coefficient. It exerts a strong control on the upper troposphere relative humidity and cloud cover (see section 3 of Hourdin, Grandpeix, et al., 2013).

Once $q_c^{in, lsc}$ and α_c^{lsc} are computed, the cloud phase is distributed among liquid droplets and ice crystals according to temperature, resulting in some of the liquid droplets to be supercooled. The fraction of cloud water in the liquid-phase x_{liq} is computed as

$$x_{liq} = \left(\frac{T - T_{min}}{T_{max} - T_{min}}\right)^n,\tag{13}$$

where T_{min} , T_{max} , and *n* were set in Version 6A to -30° C, 0° C, and 0.5 respectively. As can be seen in Figure 3, the proportion of supercooled droplets was increased in LMDZ6A to be more consistent with the most recent satellite observations (Cesana & Chepfer, 2013; Cesana et al., 2015; Choi et al., 2014; Doutriaux-Boucher & Quaas, 2004).

2.7.3. Step 3: Autoconversion

Part of the cloud water is converted to precipitation, depending on cloud phase. For liquid clouds, this corresponds to a sink term that can be written as



Figure 3. Liquid fraction x_{liq} as a function of temperature used in Versions 5A and 6A of LMDZ.

$$\frac{dq_l}{dt} = -\frac{q_l}{\tau_{conv}} \left(1 - e^{-\left(\frac{q_l/\alpha_c}{q_{chv}}\right)^2} \right),\tag{14}$$

where τ_{conv} is an autoconversion time constant and q_{clw} is a threshold condensed water amount above which autoconversion sharply increases. Note that in Equation 14, q_l is the liquid water mass mixing ratio within the gridbox, and that q_l/α_c is therefore the in-cloud liquid water content q_c^{in} . For ice clouds, the corresponding sink term follows:

$$\frac{dq_i}{dt} = \frac{1}{\rho} \frac{\partial}{\partial z} (\rho w_{iw} q_i), \tag{15}$$

where q_i is the water ice mass mixing ratio within the gridbox and where $w_{iw} = \gamma_{iw} w_0$. The fall velocity w_{iw} depends on γ_{iw} which is widely used as a tuning parameter of climate models (Hourdin et al., 2017; Mauritsen et al., 2012). The terminal fall velocity is computed according to $w_0 = 3.29(\rho q_i)^{0.16}$ (Heymsfield, 1977; Heymsfield & Donner, 1990), and depends on the mass of cloud ice without taking into account any actual size or shape of the particles. The conversion from cloud water to liquid or solid precipitation is done using a subtime step 5 times smaller than the physics time step. It is worth adding that in both



LMDZ5A and LMDZ6A, the cloud water content provided by the large-scale condensation scheme to the radiative transfer scheme is not what remains in the cloud at the end of the time step, but a mean cloud water content over the duration of the physics time step. Therefore, part of the cloud water that is converted to precipitation during the physics time step is "seen" by the radiative transfer scheme.

In Version 6A, the latent heat exchange due to the liquid \leftrightarrow ice phase change is not only implemented in the deep convection scheme (see section 2.3) but also in the large-scale condensation scheme. Moreover, when supercooled droplets are converted to precipitation, they freeze instantly, which was not the case in Version 5A. When freezing, rain releases latent heat, which can potentially bring the temperature back to above freezing. If this is the case, a small amount of rain remains liquid to stay below freezing. At the end of the large-scale condensation scheme, both the water vapor content q_v and amount of condensates $q_{l,i}$ are known, as well as the in-cloud water content $q_c^{in, lsc}$ and cloud fraction α_c^{lsc} provided by either the bi-Gaussian PDF used for shallow convection or generalized lognormal PDF used for large-scale condensation. The prognostic variables are ready for advection by the dynamical core, and the cloud water contents and fractions can be used by the radiative transfer scheme for heating rate calculations.

2.8. Radiative Transfer

Once the two cloud fractions α_c^{cv} and α_c^{lsc} are known, the total cloud fraction is estimated using

$$\alpha_c = \min(\alpha_c^{cv} + \alpha_c^{lsc}, 1), \tag{16}$$

where α_c^{lsc} includes both the cloud fraction coming from shallow convective clouds (bi-Gaussian PDF) and large-scale clouds (lognormal PDF), and where α_c^{cv} is the cloud fraction computed by the deep convection scheme. Similarly, the mean gridbox-averaged condensed water can be written as

$$q_{rad} = q_c^{in, cv} \alpha_c^{cv} + q_c^{in, lsc} \alpha_c^{lsc}.$$
(17)

 q_{rad} is used in the radiative transfer scheme to compute the optical depth and α_c is used to weight clear-sky and cloudy heating rates (precipitation is not radiatively active in LMDZ). The radiative transfer scheme uses the maximum random overlap assumption (Hogan & Illingworth, 2000; Morcrette & Fouquart, 1986). Cloud phase is determined using Equation 13. For liquid droplets, number concentration is parameterized using a modified version of Boucher and Lohmann (1995):

$$\text{CDNC} = 10^{1.3 + 0.2\log(m_{aer})},$$
 (18)

where CDNC is the cloud droplet number concentration and m_{aer} the soluble aerosol mass (instead of the sulfate aerosol mass used in Boucher & Lohmann, 1995, Equation D). Droplet sizes are then computed following Equations 2 and 4 of Boucher and Lohmann (1995). For ice crystals, particle sizes are parameterized following Equation 6 of Iacobellis and Somerville (2000) and vary in radius from r_{min} at T < -81.4°C to 61 µm at 0°C (Heymsfield, 1986), where r_{min} is a tuning parameter that varies between 3.5 and 20 µm. Note that aerosols have an impact on the size of the droplets but not on the size of the ice crystals. The first indirect effect of aerosols is therefore represented through the aerosol-dependent size of the droplets only. Liquid cloud radiative properties follow Fouquart (1988) and Smith and Shi (1992) in the SW and LW domain, respectively. Ice cloud radiative properties both in SW and LW domains are computed according to Ebert and Curry (1992). Aerosol radiative properties are computed as described in Lurton et al. (2020). LMDZ5A uses the Fouquart and Bonnel (1980) radiative transfer scheme in the SW (two bands) and LW domains (Morcrette, 1991), whereas LMDZ6A uses Fouquart and Bonnel (1980) only in the SW domain (and with six bands) and RRTM in the LW domain (Mlawer et al., 1997).

2.9. Summary of the Main Improvements

The changes affecting clouds made in Version 6A compared to Version 5A are therefore abundant and can be summarized as follows:

- New scheme for local turbulent mixing (section 2.2);
- New shallow convection scheme based on the so-called eddy-diffusivity-mass flux (EDMF) approach, coupled with the deep convection scheme; use of an improved statistical cloud scheme and



bi-Gaussian PDF of the subgrid-scale distribution of the saturation deficit; new detrainment formulation (section 2.6);

- New deep convection scheme that includes an improved mixing representation, new closure and a stochastic formulation of deep convection triggering (section 2.3);
- New parameterization of cold pools coupled with the deep convection scheme; splitting technique applied to the grid cell to distinguish the cold pool region from its environment, and to allow both the shallow and deep convections to develop outside the cold pool region (section 2.5);
- New vertical profile of the lognormal distribution's variance used for large-scale clouds (Figure 2);
- Inclusion of the latent heat exchange due to the liquid ↔ ice phase change in both the deep convection and large-scale condensation schemes;
- New formulation of the subgrid-scale rain reevaporation rate (Figure 1);
- New phase-partitioning in mixed-phase clouds (Figure 3);
- New radiative transfer scheme (section 2.8).

2.10. Lessons Learned From the Development of LMDZ6A 2.10.1. Pros and Cons of a Multiobject Framework

One of the most important aspects of LMDZ6A is the interplay between the different cloud parameterizations, that is, the shallow convection scheme, the deep convection scheme, and the so-called large-scale condensation scheme. The deep convection scheme forms a set of interconnected parameterizations that includes mixing, microphysics and thermodynamics. The representation of shallow convective clouds comes from two parameterizations, the thermal plume model and the large-scale condensation scheme. The thermal plume model transfers water from the surface to the cloud layer and provides the parameters of the subgrid-scale bi-Gaussian water distribution. The large-scale condensation scheme handles cloud formation and computes the reevaporation and autoconversion processes inside this newly formed cloud. This whole framework allows us to split into pieces complex processes and gradually link them together. It also enables the coupling between the different parameterizations, for example the deep convection triggering by thermals and cold pools (for more context on the state of the art of deep convection schemes, see Rio et al., 2019). However, each scheme provides its own cloud PDF, and ensuring a smooth transition between the different cloud PDFs is sometimes a difficult task.

2.10.2. Importance of Splitting the Grid Cell Into Two Regions

One key technical step was also distinguishing temperature and humidity inside and outside the cold pool region in both the shallow and deep convection schemes, so that both schemes run outside the cold pool region, in a more unstable environment than that of the mean atmospheric column. In Version 6A, both the thermal plumes and the deep convective updrafts thus develop in a same environment of given temperature and humidity, instead of using the mean grid cell values. Applying this splitting technique not only to the deep convection scheme (as was the case in some intermediate versions of the model) but also to the thermal plume model led to a strengthening of shallow convection relative to deep convection, and resulted in a major improvement in rainfall variability over tropical oceans. It also prevented the inhibition of shallow convection by deep convection, and that of deep convection by downdrafts and cold pools. This concept of splitting the atmospheric column in different subcolumns might be extended, in the future, to the boundary layer turbulence scheme and large-scale condensation scheme. It would allow the processes to affect temperature and humidity differently in the cloudy and clear portions of the cells. Adjusting the reevaporation rate was an essential part of the development of LMDZ6A. This rate is based on the fraction of overlying clouds (see Step 1 of section 2.7) but still affects the humidity of the whole gridbox. This splitting technique would make it possible to reevaporate rain only in the cloudy portion of the cell.

2.10.3. Revisiting Basic Thermodynamics

The development of LMDZ6A also revealed the importance of a consistent thermodynamics by the implementation of the heat exchange due to the liquid \leftrightarrow ice phase change and resulting changes in the entire cloud distribution. A disadvantage of a multiobject framework is the difficulty in ensuring thermodynamical consistency and energy conservation in the three different schemes.

2.10.4. Tuning as a Tool for Identifying Model Weaknesses

Finally, one essential lesson learned during the development of Version 6A is the need to tune the free parameters of the cloud schemes using well identified radiative targets. Beyond the technical need to tune climate models, tuning helps improve the physical formulations and identify model deficiencies "if parameter values



Table 2

Model Configurations Used in the Present Study

	LMDZ5A	LMDZ6A
Horizontal resolution Vertical resolution Run duration Physics time step	144 × 142 39 levels 20 years	144 × 142 79 levels 20 years
Boundary and initial conditions Coupling with soil model	AMIP ^a ORCHIDEE 11 layers	AMIP ^a ORCHIDEE 11 layers

^aUses observed sea surface temperatures and sea ice concentration as lower boundary condition.

needed to satisfy a given metric are outside the acceptable range, or if different values are needed for different regions or climate regimes" (Hourdin et al., 2017). We will later see, for example, that the tuning of Version 6A revealed a probable deficiency in the computation of high-level cloud cover and associated overlap assumptions (see section 5). The tuning process is also a good way to reveal compensating errors.

3. Model Setup and Evaluation

The impact of the physics improvements described in section 2 on the cloud structure and properties is analyzed using two 20 year AMIP-typed simulations that are described in Table 2. We focus on the differences

between Versions 5A and 6A of LMDZ, or more specifically between the atmospheric components of the IPSL-CM5A-MR and IPSL-CM6A-LR models, which share the same horizontal grid (144 × 142). However, we don't compare the two versions on the same vertical grid because the vertical resolution is strongly tied to the physical parameterizations of each version (39 levels for Version 5A and 79 levels for Version 6A). Thanks to the backward compatibility of LMDZ (Hourdin et al., 2020), the two simulations are run using the same source code, but the simulation is configured with LMDZ5A parameterizations in one case, and LMDZ6A parameterizations in the other. Version 5B is not analyzed in this paper because it was in many respects a prototype of Version 6A, as mentioned in section 1. The same aerosol concentration is used in both simulations, and is the one used for the CMIP6 project (Lurton et al., 2020). Both simulations are run using the most recent version of the ORCHIDEE soil and vegetation scheme. This scheme computes the vertical water transport in the soil using the Richard's equation (de Rosnay et al., 2002; D'Orgeval et al., 2008) discretized with 11 layers (see Cheruy et al., 2020, for more details on the scheme and its impact on the results of IPSL-CM6A).

The two simulations are tuned, meaning that some cloud parameters are adjusted (Hourdin et al., 2017). The tuning of LMDZ5A is described in section 3.4 of Hourdin, Grandpeix, et al. (2013), and the tuning of LMDZ6A is presented in Hourdin et al. (2020). When comparing the two simulations of the present paper, it is therefore important to keep in mind that the two simulations are tuned by targeting in particular a good TOA (top of atmosphere) global net flux. Some terms introduced in section 2 differ between LMDZ5A and LMDZ6A: ξ_{300} in Figure 2, β in Equation 12, τ_{conv} and q_{clw} in Equation 14, γ_{iw} in Equation 15 and r_{min} (the smallest ice particle size) in section 2.8. The maximum precipitation efficiency for deep convection ep_{max} is the same in the two simulations. The different values used for these parameters are summarized in Table 3. The role of each parameter in the tuning process is described in detail in Hourdin, Grandpeix, et al. (2013) and can be summarized as follows. Increasing β , τ_{conv} or q_{chv} tends to increase the amount of low-level clouds but impacts differently on their vertical profile. Increasing r_{min} decreases the emissivity of high-level clouds. Increasing the γ_{iw} coefficient increases the conversion to precipitation in ice clouds and decreases their water content. Increasing $e_{p_{max}}$ decreases the amount of detrained water and high-level clouds in convective regions. As mentioned in section 2.7, the ξ_{300} parameter has a strong impact on the relative humidity in the tropical upper troposphere and controls the variance of the lognormal PDF used in the cloud statistical scheme of high-level clouds. The latter three parameters (γ_{iw} , ep_{max} , and ξ_{300}) all affect the relative humidity of the tropical upper

troposphere as they impact on the sources (ep_{max}) and sinks $(\gamma_{iw} \text{ and } \xi_{300})$ of water vapor.

Since the two simulations are tuned, both simulations correspond to the same mean climate state. Therefore, differences between the two simulations mainly arise from changes in the model parameterizations, and to a lesser extent from slight changes in the values of the tuning parameters themselves. The impact of the physics time step and the vertical resolution were also assessed using sensitivity experiments. Changing the physics time step from 30 to 15 min in Version 5A has almost no impact on the results. The vertical resolution, however, has a noticeable impact on the results (as also noticed in other models, e.g., Xie et al., 2018), and changing the number of vertical levels from 39 to 79 levels in Version 6A increases

Table	3
-------	---

Tuning	Parameters	Used	in	the	Two	Model	Configurations	Outlined
in Table	2							

Tuning peremotor	IMD75A	I MD76A
i uning parameter	LMDZJA	LWDZ0A
ξ_{300} β ((kg m ⁻² s ⁻¹) ^{-1/2} m ⁻¹ ,	(see Figure 2) 2×10^{-5}	(see Figure 2) 1×10^{-4}
see Equation 12)		
τ_{conv} (s, see Equation 14)	1,800	900
q_{clw} (g kg ⁻¹ , see Equation 14)	0.416	0.65
γ_{iw} (see Equation 15)	0.5	0.8
r_{min} (µm, see section 2.8)	3.5	16
ep_{max} (see section 2.3)	0.999	0.999



the trade wind cumulus cloud cover by around 20% and the mid-level cloud cover in the ITCZ by around 60%.

4. Results

4.1. Cloud Spatial Distribution

We first compare the simulated cloud distribution to the lidar-based GOCCP data set (GCM Oriented CALIPSO Cloud Product, Chepfer et al., 2010). To do so, the cloud water contents and fractions predicted by LMDZ are processed by the CALIPSO-COSP simulator (Bodas-Salcedo et al., 2011) to derive the cloud fractions and covers the instrument would see if it was observing the model. To do so, the simulator uses the same overlap assumption as the LMDZ radiative transfer. *Note that in the present paper, the term "cloud fraction" refers to the 3-D cloud fraction at each level and in each gridbox, whereas the term "cloud cover" refers to the total cloud cover seen from above, computed by integrating the 3-D cloud fractions vertically assuming a given overlap of clouds within the vertical column of the model gridboxes. This integral can be over the entire column or over*



Figure 4. (a–i) Low, middle, and high cloud cover from the CALIPSO-GOCCP climatology (averaged over the 2006–2009 period, top row) and from Versions 6A and 5A of LMDZ (as computed by the CALIPSO-COSP simulator and averaged over a 20 year period). (j–o) Difference between the simulated cloud covers and the CALIPSO-GOCCP climatology. A positive value implies overestimation of the cloud cover by the model.





Figure 5. Zonally averaged vertical structure of the cloud fractions predicted by LMDZ5A and 6A (20 year average) using the COSP simulator (d–i) compared against the CALIPSO-GOCCP climatology (a–c). *Y* axis gives the altitude above the local surface in km. The dotted and solid white contours represent the 0.05 and 0.1 cloud fractions, respectively. The left column gives the total cloud fraction, the middle column the liquid-phase cloud fraction, and the right column the ice phase cloud fraction.

a given pressure interval. In our case, we use three cloud covers that correspond to three cloud categories: low-level clouds (below 680 hPa or \sim 3 km), mid-level clouds (between 680 and 440 hPa, that is, 3 and 6.5 km) and high-level clouds (above 440 hPa or \sim 6.5 km). Figure 4 shows the cloud cover maps and bias maps of the three cloud categories, whereas Figure 5 shows the 3-D cloud fractions. Table 4 also summarizes the mean bias between the model and the observations, the RMSE and the correlation coefficient.

Starting with low-level clouds, comparing Figures 4a, 4d, and 4g reveals a significant improvement in the low-level cloud covers over the tropical oceans in LMDZ6A. On the west side of ocean basins, trade wind cumulus clouds were underestimated in Version 5A, as can be seen in Figures 4a and 4g. In LMDZ6A, they

Table 4

Mean Bias, Root-Mean-Square Error, and Correlation Coefficient for Low-Level, Mid-level, and High-Level Cloud Covers Between Both Versions of the Model and the CALIPSO-GOCCP Climatology

	Low-level		Cloud level Mid-level		High-level	
Model version	5A	6A	5A	6A	5A	6A
Mean bias RMSE Correlation coefficient	-0.106 0.156 0.829	-0.006 0.119 0.840	-0.144 0.163 0.543	-0.035 0.059 0.741	0.030 0.089 0.628	-0.122 0.137 0.758
	-					

Note. See Figure 4 for context.

reach a better agreement with the observations (see Figures 4a and 4g. In EMD201, inc) reach a better agreement with the observations (see Figures 4a and 4d). On the east side of ocean basins, stratocumulus clouds are improved in LMDZ6A due to the new statistical cloud scheme and change in the detrainment formulation of the thermal plume model (see section 2.6). Low-level clouds were underestimated over the Indo-Pacific warm pool in LMDZ5A and are now better represented as well. As can be seen in the bias plots Figures 4j and 4m, the overall bias is reduced in Version 6A over the tropical oceans but stratocumulus cloud cover maxima are slightly shifted away from the coasts. As described in Hourdin et al. (2019), this shift might be due to the tendency of the LMDZ6A model to maintain a 100% cloud deck for too long during the transition from stratocumulus to cumulus clouds. Outside the tropical belt, low-level clouds are overestimated over the Arctic and Southern Oceans. As evidenced in



Table 5

Mean Bias, Root-Mean-Square Error, and Correlation Coefficient for SW and LW CRE Between Both Versions of the Model and the CERES Observations (Loeb et al., 2009)

	CRE wavelength range			
	Short	wave	Longwave	
Model version	5A	6A	5A	6A
Mean bias	-5.043	-0.795	5.932	-0.818
RMSE	14.916	9.150	9.224	4.630
Correlation coefficient	0.827	0.881	0.708	0.855

Note. See Figure 7 for context.

Figures 4j and 4m, this bias is stronger in Version 6A than in Version 5A. The overall RMSE for low-level clouds is reduced in Version 6A (see Table 5), mostly thanks to the improvements seen in the tropical regions.

The mid-level cloud distribution is one of the most striking improvements of LMDZ6A. A comparison of Figure 4b and 4e shows a reasonable agreement between the model and the observations, whereas previous versions of the model were systematically underestimating mid-level clouds. This is due to the improvement of the deep and shallow convection schemes in the tropical and mid-latitude regions (see sections 2.3 and 2.6), and to the new phase-partitioning of clouds in the mid-latitude to high-latitude regions (see Figure 3). As mentioned in section 3, the increase in vertical

resolution from 39 levels in Version 5A to 79 levels in Version 6A also improved mid-level cloud covers in the ITCZ. High-level clouds are however underestimated in LMDZ6A, which was not the case before (Figure 4, right column). We had to reach a compromise in the tuning of the fall velocity parameter γ_{iw} , which is relatively high in Version 6A (see Table 3). This tends to reduce the amount of high-level clouds globally to meet the LW CRE tuning target.

Figure 5 shows the zonal mean cloud fractions averaged over 20 years of simulation in the two versions of the model and in the CALIPSO-GOCCP data set. As already noticed in Figure 4, outside the tropical belt, low-level clouds are overestimated in both LMDZ5A and 6A, but their altitude and fraction are improved in LMDZ6A. Their altitude of around 2 km is now slightly too high compared to the observations where low-level clouds are mostly below 1.5 km. Interestingly, comparing Figure 5e and 5h reveals that in Version 6A, we actually decreased the 3-D cloud fraction, but increased the geometrical thickness of low-level clouds, thereby increasing the low-level cloud cover (see Figure 4d). Mid-level clouds were mostly absent in LMDZ5A and are now better represented (see Figure 5e and 5h), especially over mid-latitude to high-latitude regions. This is also evidenced by the mean bias, RMSE and correlation coefficient shown in Table 4. In the tropics, LMDZ6A shows a local maximum in mid-level cloud cover slightly below 5 km altitude. The same maximum is located more than a 1,000 m higher in the observations, at elevations devoid of any cloud in the model. Another striking improvement of Version 6A is the water phase-partitioning in mid-level to high-level clouds. In LMDZ5A, the ice phase cloud fraction was clearly overestimated (see Figure 5i) and not consistent with the observations (Cesana et al., 2015). Changing the phase-partitioning in mixed-phase clouds (as shown in Figure 3) significantly improved the ice phase cloud fractions in LMDZ6A (Figure 5, right column), as well as the liquid-phase cloud fractions in mid-level clouds (middle column). As previously mentioned, high-level cloud cover remains underestimated due to a compromise in the tuning of the model, but their spatial distribution is improved (see Figure 4f and correlation coefficients in Table 4). High-level 3-D cloud fractions are overestimated in the tropical regions if we compare Figures 5c and 5f, but their total column cloud cover is underestimated in this same region if we look at Figure 4f and upper left panel of Figure 8. This suggests, as will be discussed in section 5, that the cloud cover computed by the model for high-level clouds is too low and compensated by a too high 3-D cloud fraction.

Figure 6 focuses on the cloud fraction in the tropical regions, more exactly on the GPCI transect, which spans from San Francisco to Honolulu (see Teixeira et al., 2011, for more details). This transect is especially useful



Figure 6. Cross section of the cloud fraction along the GPCI transect (GCSS/WGNE Pacific Cross-Section Intercomparison Teixeira et al., 2011) as observed by CALIPSO-GOCCP over the 2006–2009 period (left panel) and simulated by LMDZ5A (middle panel) and LMDZ6A (right panel) over a 20 year period.







Figure 7. (a–f) Shortwave (left column) and longwave (right column) cloud radiative effect (CRE) in W m⁻² observed by CERES (averaged over a 16 year period Loeb et al., 2009) and simulated by Version 6A (c and d) and Version 5A (e and f) of the LMDZ climate model (averaged over a 20 year period). To make use of a common color scale, the opposite of the SW CRE is represented: A positive value thus corresponds to an increased reflection and decrease in the amount of solar radiation absorbed by the Earth relative to clear-sky conditions. (g–j) bias plots for Version 6A (g and h) and Version 5A (i and j). A positive value of the SW CRE bias implies overestimation of the SW CRE by the model (not enough reflection by clouds), and a positive value of the LW CRE bias implies overestimation of the LW CRE by the model (too much greenhouse effect of clouds).

to evaluate the representation of the stratocumulus to cumulus (Sc-to-Cu) and shallow to deep convection transitions in climate models. In LMDZ5A, the Sc-to-Cu transition was visible, but stratocumulus clouds were too close to the surface and high-level cloud fractions were overestimated. Version 6A nicely represents the Sc-to-Cu transition and shows a better evolution of the cloudy boundary layer, but clouds tend to extend beyond the 2 km height seen in the observations. Over the warmer waters of the trade wind boundary layer (around 5°N), the model cloud fractions remain too low compared to the observations. Mid-level cloud fractions are also underestimated in deep convective regimes, as is also noticed in Figure 5d at around 8 km altitude. This altitude range is where the $\xi(p)$ function sharply increases (see Figure 2). It is therefore in the





Figure 8. (left column) Zonal mean cloud covers simulated by LMDZ5A and 6A (20 year average) using the COSP simulator, compared against the CALIPSO-GOCCP climatology (in gray). (right column) Zonal mean TOA (top of atmosphere) SW (top panel), LW (middle panel), and total (lower panel) CRE (cloud radiative effect) predicted by LMDZ5A and 6A (20 year average) and observed by the CERES instruments (EBAF data set Loeb et al., 2009).

transition zone between the PDFs of the shallow convection, deep convection and large-scale condensation schemes, and suggests that the interplay between the schemes need to be improved in this region. The high-level cloud fraction is better represented in LMDZ6A but clouds remain too geometrically thin compared to the observations.

4.2. Cloud Radiative Effect

Clouds play a crucial role in the radiative budget of the atmosphere, and a compromise has often to be found between a good representation of their properties and a good TOA energy budget of the model. The tuning method of LMDZ6A is described in Hourdin et al. (2020), and we focus here on the role of clouds in the radiative budget.

Figure 7 shows the observed and simulated Cloud Radiative Effect (CRE) in the SW and LW domains, as well as the bias maps. The left column of this figure shows a clear improvement of the SW CRE, especially in mid-latitude to high-latitude regions where reflection by low-level clouds was too high in Version 5A. This improvement results in a 5 W m^{-2} reduction of the SW CRE mean bias and RMSE in LMDZ6A, as shown in Table 5. An improvement of the same magnitude is seen in the LW CRE, but in this latter case, the spatial distribution is also improved (see the increase in the correlation coefficient in Table 5), which is less the case of the SW CRE, especially in the tropical regions. Indeed, despite a clear improvement of the SW CRE in the ITCZ (see Figure 7c), the SW radiative effect of stratocumulus clouds is shifted away from the coast over the eastern part of tropical ocean basins, and trade wind cumulus clouds reflect less sunlight than in the observations (see Figure 7g). These biases are consistent with those of the low-level cloud cover described in section 4.1.





Figure 9. Cloud LWP and IWP (liquid and ice water paths in $g m^{-2}$) over oceans in LMDZ5A (top) and LMDZ6A (bottom).

The left column of Figure 8 summarizes the zonal mean cloud cover of the three cloud categories and the corresponding radiative forcings in the right column. In the tropics, cloud covers are improved in LMDZ6A at all levels, but remain slightly lower than in the observations. The right column of Figure 8 shows that in this region a realistic CRE is reached even though the cloud covers are slightly biased low. For high-level clouds, this suggests that the underestimated cloud covers are probably compensated by a too high 3-D cloud fraction. For low-level clouds, it suggests that the underestimated cloud cover is compensated by overly bright low-level clouds, as will be discussed in section 5. The situation is different over the Arctic and Southern oceans, where a realistic CRE is reached even though the low-level cloud covers are biased high (see Figures 7c, 7d, and 8, lower left panel). In these regions, the LMDZ6A SW CRE is in better agreement with the observations than that of LMDZ5A, and this has to do with cloud phase and opacity, as we will see in the next paragraph. It is worth noting that this difference in low-level cloud covers between the two versions could have come from the results of the simulator because of the possible screening of low-level clouds by high-level clouds. In our case, high-level cloud covers are biased low relative to the observations in Version 6A (see Figure 4l) and could increase the signal coming from low-level clouds and partly explain the positive cloud cover bias seen in Figure 4d. But this is not the case. We find the same difference between the low-level cloud covers of Versions 5A and 6A using the results of the model radiative transfer itself (not the simulator).

Let's now return to the good total CRE simulated in LMDZ6A in mid-latitude to high-latitude regions despite the biases seen in the various cloud covers (Figure 8). In mid-latitude to high-latitude regions, phase-partitioning has been found to be strongly connected to the SW CRE in many models (McCoy et al., 2016). In our case, sensitivity experiments show that increasing the temperature range of supercooled droplets leads to a greater vertical extension of liquid clouds, which are otherwise confined to lower layers. This results in a higher concentration, in LMDZ6A, of liquid droplets in mid-level clouds, where droplets are more reflective than ice (Liou, 2003), but more importantly in a lower concentration of droplets in low-level clouds. This decrease in the concentration of liquid droplets in low-level clouds explains why the SW CRE is in better agreement with the observations in LMDZ6A, despite the overestimation of the low-level cloud cover. The LW CRE is also sensitive to phase-partitioning in mixed-phase clouds. The left column of Figure 8 shows that LMDZ6A has less high-level clouds cover decreases the LW CRE, but on the other hand, the increase in mid-level cloud covers of high liquid content strongly increases it. In the end, the LW CRE in LMDZ6A is reduced by the right amount compared to that of LMDZ5A and is in good agreement with the





Figure 10. Regime sorted plots of the SW (top), LW (middle), and net (bottom) CRE as a function of ω 500 in hPa/day between 30°S and 30°N and over the oceans. For comparison, the black line shows the same diagnostics obtained using ERA reanalysis and the CERES data (EBAF data set Loeb et al., 2009).

observations. The overall cloud liquid water path in mid-latitude to high-latitude regions is increased, as illustrated in Figure 9, whereas the ice water path is strongly decreased. Satellite retrieval of the LWP and IWP is not an easy task, but a comparaison of the simulated LWP with the work of O'Dell et al. (2008) suggests that it is in good agreement with the observations in the tropical regions and slightly too high in mid-latitude to high-latitude regions. It is more difficult to compare the simulated cloud IWP (i.e., the nonprecipitating ice) to existing observations, but the sharp decrease in the cloud IWP of Version 6A is more in line with the cloud IWP found in other models, including the ERA5 and MERRA-2 reanalyses (see Figure 3 of Duncan & Eriksson, 2018).

Figure 10 focuses on the tropics and shows the simulated CRE as a function of the dynamical regimes (through the vertical velocity ω at 500 hPa). This type of analysis, introduced by Bony et al. (2004), shows how well the CRE is represented in regions of subsidence ($\omega > 0$) and updraft ($\omega < 0$). Both the SW and LW CREs show a gradual decrease (in terms of absolute value for the SW CRE) from regions of strong updrafts where clouds are abundant to regions of strong subsidence where clouds dissipate. The lower panel of Figure 10 shows a clear improvement of the total CRE in LMDZ6A in convective regions ($\omega <$ 0). This is mostly due to an improvement of the SW CRE (upper panel), and reflects the changes applied to the thermal plume parameterization, which improved both the stratocumulus clouds over the eastern part of tropical ocean basins and trade wind cumulus clouds (see section 4.1). However, the SW and LW CREs are still too weak in magnitude in strongly convective regions ($\omega < -40 \text{ hPa/day}$) and the SW CRE is higher than observed in regions of strong subsidence ($\omega > 20$ hPa/day).

5. Discussion

Thanks to the improvements of the physical parameterizations and to an experience gained in the tuning of the model, the cloud distribution and radiative effects have been significantly improved in LMDZ6A. But the refined tuning of the model has also underlined structural problems, especially in the detailed cloud radiative properties. In particular, the difficulty to tune high-level clouds points to an inappropriate representation of their radiative properties, which impacts on all clouds. Difficulties in modeling the properties of high-level clouds were already found in the early versions of LMDZ (Webb et al., 2001). Figure 11 shows the PDF of the high-level cloud cover over the tropical oceans based on the daily outputs of the CALIPSO-GOCCP observations (left panel) and results of the LMDZ model simulator (middle and right panels). The observed PDF is a highly skewed-right distribution with a peak at 0-5% cloud cover and an outlier at 97.5-100%. The LMDZ5A PDF shows a lower peak at 0-5% but an otherwise similar distribution, with a smaller outlier at 97.5-100%. LMDZ6A shows a skewed-right distribution similar to the observations for cloud cover lower than 20%, but its PDF differs significantly for

higher cloud covers, with a strong decrease above 50% and no outlier at 97.5–100%. This difficulty of LMDZ6A to attain complete coverage for high-level clouds might explain why these clouds are hard to tune in this version. Therefore, work is underway to improve the ξ function (see section 2.7) using a more physical parameterization, as well as the overlap assumptions and subgrid-scale heterogeneities of high-level clouds.

Regarding tropical low-level clouds, Figure 12 shows the density of points of a given cloud reflectance and cloud cover in the observations (left panel) and in the model (middle and right panels, see Konsta





Figure 11. PDF of the high-level cloud cover over the tropical oceans. (left panel) CALIPSO-GOCCP daily observations over the 2007–2008 period. (middle and right panels) Daily cloud covers computed by the CALIPSO-COSP simulator in Versions 5A and 6A of LMDZ over a 10 year period.

et al., 2016, for more details on the method). Cloud reflectance in Figure 12 is a function of the vertically integrated cloud optical depth, whereas cloud cover will be more dependent on the cloud fraction vertical profiles and overlap assumption. Two populations can be identified in the observations (Figure 12, left panel): Trade wind cumulus clouds have low reflectance and cover values, whereas stratocumulus clouds have medium reflectance and high cover values. The observations also show an increase in cloud reflectance with increasing cloud cover. LMDZ5A was showing the opposite tendency (Figure 12, middle panel) and trade wind cumulus clouds were too bright in this version of the model, a problem commonly referred to as the "too few, too bright" problem (Nam et al., 2012). As explained in Konsta et al. (2016), this increase in reflectance with decreasing cloud fraction in LMDZ5A was due to the activation of the deep convection scheme in trade wind regions, which affected the low-level cloud PDFs. The implementation of the thermal plume model in LMDZ6A clearly improved the distribution, which is now closer to the observations (Figure 12, right panel). However, in LMDZ6A, trade wind cumulus clouds are still too reflective and their cover is too low. Stratocumulus clouds are well represented and show medium reflectance and high cover values, in agreement with the observations. Between these two populations, a third population appears in the model, and is characterized by cloud reflectance values of around 0.2 and cover values between 0.6 and 0.9. The too few, too bright bias was thus reduced but not fully solved. Despite the high number in LMDZ6A of low cloud cover values compared to the observations (Figure 12, right panel), the SW CRE is still in good agreement with the observations. This suggests that this too low cover is compensated by an excessive brightness in the tuning process, which targets the CRE as a



Figure 12. 2-D histograms of low-level cloud reflectances and covers over the tropical oceans (30°S to 30°N) observed by PARASOL and CALIPSO-GOCCP (left panel) and simulated by LMDZ5A (middle panel) and LMDZ6A (right panel) using instantaneous outputs (Konsta et al., 2016).



priority. We thus see cloud reflectances of around 0.3 in LMDZ6A, compared to less than 0.1 in the observations (see Figure 12, left and right panels). This shows the limit of the maximum random overlap assumption used in LMDZ6A. Preliminary sensitivity experiments performed with LMDZ6A shows that using the exponential-random overlap assumption (Hogan & Illingworth, 2000) instead of the maximum random overlap assumption may improve the distribution shown in Figure 12 by increasing cumulus cloud cover. Another way to increase low-level cloud covers is to represent subgrid-scale vertical heterogeneities by distinguishing the cloud fraction by volume from the cloud fraction by surface. The latter was found to be 20% greater on average than the cloud fraction by volume (Brooks et al., 2005). The cloud fraction by surface is more appropriate for coupling with radiative transfer schemes but most climate models do not yet distinguish between the two quantities and by doing so, assume that the cloudy area of a gridbox fills the entire gridbox in the vertical. The difference between the cloud fraction by volume and the cloud fraction by surface can be computed by a parameterization of subgrid-scale heterogeneities that will depend on the vertical resolution and various physical information, such as wind shear (Sulak et al., 2020). Work is underway to implement such parameterization in LMDZ (Jouhaud et al., 2018). This could improve the CRE of low-level clouds but also high-level clouds.

6. Conclusion

After a series of parameterization changes (summarized in section 2.9) and a finer tuning of the radiative budget, several cloud features were improved in Version 6A of the LMDZ climate model:

- Low-level (below 3 km) cloud covers are improved both in trade wind regions and in the east side of ocean basins (see Figure 4d), due to the new shallow convection scheme;
- Mid-level clouds, which were almost inexistent in LMDZ5A, are much better represented in LMDZ6A (see Figure 4e). Mid-level to high-level cloud phase is also more realistic and now includes a more realistic fraction of supercooled liquid droplets (see Figure 5e). These improvements mostly come from the changes made in the deep convection scheme in the tropical regions, and in the new phase-partitioning in the midlatitude to high-latitude regions;
- CREs are improved (see Figure 8, right column) and LMDZ6A shows a 5 W m⁻² improvement in both the SW and LW CRE compared to LMDZ5A (see Table 5), due to the combined effect of the new shallow convection scheme and new phase-partitioning;
- A 20 W m⁻² bias in the SW CRE of the convective regions is corrected (see Figure 10, upper panel), thanks mostly to the new shallow convection scheme;
- Tropical low-level cloud reflectance and cover are significantly improved (see Figure 12, right panel) due to the shallow convection scheme and its new statistical cloud scheme based on a Bi-gaussian PDF.

The finer model tuning performed for LMDZ6A also revealed structural errors and inconsistencies that call for a revisit of some existing parameterizations. Indeed, the model reaches a good radiative balance for cloud covers that are sometimes strongly biased. This is true for low-level clouds but more importantly for high-level clouds, whose covers need to be lower than observed to restore the radiative balance. For clouds of all levels, work is underway to improve the overlap assumptions of the radiative transfer scheme and to better account for the cloud subgrid-scale heterogeneities (see, e.g., Jouhaud et al., 2018). High-level clouds also rely on a fixed value of the lognormal PDF variance (ξ_{300}) which must be improved and more physically based. Mid-level clouds are also the focus of current development efforts, in order to better represent the deepening of shallow cumulus clouds into congestus clouds (see Figure 6). Improvement of the cloud microphysical scheme is also underway, with a particular focus on cold and mixed-phase clouds. Priorities include the improvement of the conversion of ice clouds to solid precipitation (Lemonnier et al., 2020), the implementation of supersaturation with respect to ice (Genthon et al., 2017), and the representation of subgrid-scale processes in mixed-phase clouds.

Appendix A: Notations

ρ

- Atmospheric density (kg m^{-3})
- ω 500 Large-scale vertical velocity at 500 hPa (hPa day⁻¹)
- θ Potential temperature (K)
- q_{ν} Water vapor mass mixing ratio (kg kg⁻¹)



			1.
a	Liquid water	maga miving	rotio $(\log \log^{-1})$
u1	LIUUIU WALEI	111455 1111X1112	
11		0	

- q_i Ice mass mixing ratio (kg kg⁻¹)
- q_t Total water mass mixing ratio (kg kg⁻¹)
- q_{t_c} Gridbox mean amount of condensate and in-cloud vapor (kg kg⁻¹)
- q_{sat} Saturation mass mixing ratio (kg kg⁻¹)
- s Saturation deficit (see Equation 3 of Jam et al., 2013) (kg kg⁻¹)
- P(q) Probability Density Function (PDF) of water vapor q
- *Q*(*s*) Probability Density Function (PDF) of the saturation deficit *s*
- ALE Available Lifting Energy $(J kg^{-1})$
- ALP Available Lifting Power (W m^{-2})
- w_{iw} Fall velocity of ice crystals (m s⁻¹)
- w_0 Terminal fall velocity of ice crystals (m s⁻¹)
- $P_{l,i}$ Liquid/Ice precipitation flux density (kg m⁻² s⁻¹)
- $d\theta_{dw}^{cv}$ Temperature tendency due to downdrafts (K s⁻¹)
- $dq_{t,dw}^{cv}$ Total water tendency due to downdrafts (kg kg⁻¹ s⁻¹)
- α_{th} Coverage fraction of thermals
- θ_{env} θ in the environment of the plume (K)
- $q_{t, env}$ Mean q_t in the environment of the plumes (kg kg⁻¹)
- s_{env} Saturation deficit in the environment of the plumes (kg kg⁻¹)
- σ_{env} σ of the PDF related to the environment of the plumes (kg kg⁻¹)
- s_{th} Saturation deficit inside the plumes (kg kg⁻¹)
- σ_{th} σ of the PDF related to the plumes (kg kg⁻¹)
- q_c^{in} In-cloud water mass mixing ratio (kg kg⁻¹)
- α_c Cloud fraction
- q_m Condensed water mixing ratio in the mixed drafts (kg kg⁻¹)
- M_t Mass flux density of the mixed drafts (kg m⁻² s⁻¹)
- α_m Coverage fraction of mixed drafts
- τ_m Dissipation time constant of the saturated drafts (s)
- δz Vertical spacing of gridboxes (m)
- M_a Mass flux density of the undiluted updrafts (kg m⁻² s⁻¹)
- α_a Coverage fraction of undiluted updrafts
- w_a Vertical velocity of the undiluted updrafts (m s⁻¹)

When written in superscript, *th*, *wk*, *cv* and *lsc* indicates variables related to thermal plumes, wakes, deep convection, and large-scale condensation, respectively.

For a list of the tuning parameters and their notations, see Table 3.

Acknowledgments

J. B. M. thanks Christophe Genthon, Laurent Li, and Jean Jouhaud for useful discussions; Julie Celton-Madeleine for proofreading the manuscript; and Jérôme Servonnat and Karine Marquois for technical assistance. This study benefited from the IPSL mesocenter ESPRI facility which is supported by CNRS, Sorbonne University, Labex LIPSL, CNES, and Ecole Polytechnique. It was supported by CNRS and by CNES and granted access to the HPC resources of IDRIS under Allocation 0292 made by GENCI. It was also supported by the DEPHY2 project funded by the French national program LEFE/INSU. We finally thank the JAMES editorial team and two anonymous referees for their thoughtful comments, which very much improved the quality of the manuscript.

Data Availability Statement

The last version of the LMDZ source code can be downloaded freely on the LMDZ website (https://lmdz. lmd.jussieu.fr). The version used for the specific simulation runs of this paper is the svn release 3404 from 16 October 2018 which can be downloaded and installed on a Linux computer by running the install_lmdz.sh script available online on the LMDZ website. A large part of the outputs is available on the CMIP5 and CMIP6 archives, distributed through the Earth System Grid Federation (ESGF) and freely accessible through the ESGF data portals after registration. Details about ESGF are presented on the CMIP Panel website (at https://www.wcrp-climate.org/index.php/wgcm-cmip/about-cmip). Lighter postprocessed files together with the scripts used to generate the figures are registered under https://doi.org/10.5281/ zenodo.3942031 and available online (at https://doi.org/10.5281/zenodo.3942031).

References

Betts, A. K. (1973). Non-precipitating cumulus convection and its parameterization. Quarterly Journal of the Royal Meteorological Society, 99(419), 178–196. https://doi.org/10.1002/qj.49709941915

Bodas-Salcedo, A., Webb, M. J., Bony, S., Chepfer, H., Dufresne, J. L., Klein, S. A., et al. (2011). COSP: Satellite simulation software for model assessment. *Bulletin of the American Meteorological Society*, 92(8), 1023–1043. https://doi.org/10.1175/2011BAMS2856.1





Bony, S., Colman, R., Kattsov, V. M., Allan, R. P., Bretherton, C. S., Dufresne, J.-L., et al. (2006). How well do we understand and evaluate climate change feedback processes? *Journal of Climate*, 19, 3445. https://doi.org/10.1175/JCLI3819.1

Bony, S., Dufresne, J.-L., Le Treut, H., Morcrette, J.-J., & Senior, C. (2004). On dynamic and thermodynamic components of cloud changes. *Climate Dynamics*, 22, 71–86. https://doi.org/10.1007/s00382-003-0369-6

Bony, S., & Emanuel, K. A. (2001). A parameterization of the cloudiness associated with cumulus convection; evaluation using TOGA COARE data. Journal of Atmospheric Sciences, 58, 3158–3183. https://doi.org/10.1175/1520-0469(2001)058<3158:APOTCA>2.0.CO;2

Boucher, O., & Lohmann, U. (1995). The sulfate-CCN-cloud albedo effect. *Tellus Series B Chemical and Physical Meteorology B*, 47, 281. https://doi.org/10.1034/j.1600-0889.47.issue3.1.x

Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, 21, e2019MS002010. https://doi.org/10.1029/2019MS002010

- Brooks, M. E., Hogan, R. J., & Illingworth, A. J. (2005). Parameterizing the difference in cloud fraction defined by area and by volume as observed with radar and lidar. *Journal of Atmospheric Sciences*, 62, 2248–2260. https://doi.org/10.1175/JAS3467.1
- Cesana, G., & Chepfer, H. (2013). Evaluation of the cloud thermodynamic phase in a climate model using CALIPSO-GOCCP. Journal of Geophysical Research: Atmospheres, 118, 7922–7937. https://doi.org/10.1002/jgrd.50376

Cesana, G., Waliser, D. E., Jiang, X., & Li, J. L. F. (2015). Multimodel evaluation of cloud phase transition using satellite and reanalysis data. Journal of Geophysical Research: Atmospheres, 120, 7871–7892. https://doi.org/10.1002/2014JD022932

Chepfer, H., Bony, S., Winker, D., Cesana, G., Dufresne, J. L., Minnis, P., et al. (2010). The GCM-oriented CALIPSO cloud product (CALIPSO-GOCCP). Journal of Geophysical Research, 115, D00H16. https://doi.org/10.1029/2009JD012251

Cheruy, F., Ducharne, A., Hourdin, F., Musat, I., Vignon, E., Gastineau, G., et al. (2020). Improved near surface continental climate in IPSL-CM6A-LR by combined evolutions of atmospheric and land surface physics. *Journal of Advances in Modeling Earth Systems*, *12*, e2019MS002005. https://doi.org/10.1029/2019MS002005

Choi, Y.-S., Ho, C.-H., Park, C.-E., Storelvmo, T., & Tan, I. (2014). Influence of cloud phase composition on climate feedbacks. Journal of Geophysical Research: Atmospheres, 119, 3687–3700. https://doi.org/10.1002/2013JD020582

D'Orgeval, T., Polcher, J., & de Rosnay, P. (2008). Sensitivity of the West African hydrological cycle in ORCHIDEE to infiltration processes. Hydrology and Earth System Sciences, 12, 1387–1401. https://doi.org/10.5194/hess-12-1387-2008

de Rosnay, P., Polcher, J., Bruen, M., & Laval, K. (2002). Impact of a physically based soil water flow and soil-plant interaction representation for modeling large-scale land surface processes. *Journal of Geophysical Research*, 107(D11), 4118. https://doi.org/10.1029/ 2001JD000634

Doutriaux-Boucher, M., & Quaas, J. (2004). Evaluation of cloud thermodynamic phase parametrizations in the LMDZ GCM by using POLDER satellite data. *Geophysical Research Letters*, *31*, L06126. https://doi.org/10.1029/2003GL019095

Dufresne, J.-L., & Bony, S. (2008). An assessment of the primary sources of spread of global warming estimates from coupled atmosphere ocean models. *Journal of Climate*, 21(19), 5135. https://doi.org/10.1175/2008JCLI2239.1

Duncan, D. I., & Eriksson, P. (2018). An update on global atmospheric ice estimates from satellite observations and reanalyses. Atmospheric Chemistry & Physics, 18(15), 11,205–11,219. https://doi.org/10.5194/acp-18-11205-2018

Ebert, E. E., & Curry, J. A. (1992). A parameterization of ice cloud optical properties for climate models. *Journal of Geophysical Research*, 97(D4), 3831–3836. https://doi.org/10.1029/91JD02472

Emanuel, K. A. (1991). A scheme for representing cumulus convection in large-scale models. Journal of Atmospheric Sciences, 48(21), 2313–2329. https://doi.org/10.1175/1520-0469(1991)048<2313:ASFRCC>2.0.CO;2

Emanuel, K. A., & Ivkovi-Rothman, M. (1999). Development and evaluation of a convection scheme for use in climate models. Journal of Atmospheric Sciences, 56(11), 1766–1782. https://doi.org/10.1175/1520-0469(1999)056<1766:DAEOAC>2.0.CO;2

Eyring, V., Bony, S., Meehl, G. A., Senior, C., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2015). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organisation. *Geoscientific Model Development Discussions*, 8(12), 10,539–10,583. https://doi.org/10.5194/gmdd-8-10539-2015

Fouquart, Y. (1988). Radiative transfer in climate models. In M. E. Schlesinger (Ed.), Physically-based modelling and simulation of climate and climatic change (pp. 223–283). Dordrecht: Springer.

Fouquart, Y., & Bonnel, B. (1980). Computations of solar heating of the Earth's atmosphere—A new parameterization. *Beitraege zur Physik* der Atmosphaere, 53, 35–62.

Genthon, C., Piard, L., Vignon, E., Madeleine, J.-B., Casado, M., & Gallée, H. (2017). Atmospheric moisture supersaturation in the

near-surface atmosphere at Dome C, Antarctic Plateau. Atmospheric Chemistry and Physics, 17(1), 691–704. https://doi.org/10.5194/acp-17-691-2017

Grandpeix, J.-Y., & Lafore, J.-P. (2010). A density current parameterization coupled with Emanuel's convection Scheme. Part I: The models. Journal of Atmospheric Sciences, 67(4), 881–897. https://doi.org/10.1175/2009JAS3044.1

Grandpeix, J.-Y., Lafore, J.-P., & Cheruy, F. (2010). A density current parameterization coupled with Emanuel's convection scheme. Part II: 1D simulations. *Journal of Atmospheric Sciences*, 67(4), 898–922. https://doi.org/10.1175/2009JAS3045.1

Grandpeix, J. Y., Phillips, V., & Tailleux, R. (2004). Improved mixing representation in Emanuel's convection scheme. Quarterly Journal of the Royal Meteorological Society, 130(604), 3207–3222. https://doi.org/10.1256/qj.03.144

Heymsfield, A. J. (1977). Precipitation development in stratiform ice clouds: A microphysical and dynamical study. Journal of Atmospheric Sciences, 34, 367–381. https://doi.org/10.1175/1520-0469(1977)034<0367:PDISIC>2.0.CO;2

Heymsfield, A. J. (1986). Ice particles observed in a cirriform cloud at -83°C and implications for polar stratospheric clouds. *Journal of Atmospheric Sciences*, 43(8), 851-856. https://doi.org/10.1175/1520-0469(1986)043<0851:IPOIAC>2.0.CO;2

Heymsfield, A. J., & Donner, L. J. (1990). A scheme for parameterizing ice-cloud water content in general circulation models. *Journal of Atmospheric Sciences*, 47, 1865–1877. https://doi.org/10.1175/1520-0469(1990)047<1865:ASFPIC>2.0.CO;2

Hogan, R. J., & Illingworth, A. J. (2000). Deriving cloud overlap statistics from radar. Quarterly Journal of the Royal Meteorological Society, 126(569), 2903–2909. https://doi.org/10.1002/qj.49712656914

Hourdin, F., Couvreux, F., & Menut, L. (2002). Parameterization of the dry convective boundary layer based on a mass flux representation of thermals. *Journal of Atmospheric Sciences*, 59(6), 1105–1123. https://doi.org/10.1175/1520-0469(2002)059<1105:POTDCB>2.0.CO;2

Hourdin, F., Foujols, M.-A., Codron, F., Guemas, V., Dufresne, J.-L., Bony, S., et al. (2013). Impact of the LMDZ atmospheric grid configuration on the climate and sensitivity of the IPSL-CM5A coupled model. *Climate Dynamics*, 40(9–10), 2167–2192. https://doi.org/ 10.1007/s00382-012-1411-3

Hourdin, F., Grandpeix, J.-Y., Rio, C., Bony, S., Jam, A., Cheruy, F., et al. (2013). LMDZ5B: The atmospheric component of the IPSL climate model with revisited parameterizations for clouds and convection. *Climate Dynamics*, 40, 2193–2222. https://doi.org/10.1007/s00382-012-1343-y



- Hourdin, F., Jam, A., Rio, C., Couvreux, F., Sandu, I., Lefebvre, M.-P., et al. (2019). Unified parameterization of convective boundary layer transport and clouds with the thermal plume model. *Journal of Advances in Modeling Earth Systems*, *11*, 2910–2933. https://doi.org/ 10.1029/2019MS001666
- Hourdin, F., Mauritsen, T., Gettelman, A., Golaz, J.-C., Balaji, V., Duan, Q., et al. (2017). The art and science of climate model tuning. Bulletin of the American Meteorological Society, 98(3), 589–602. https://doi.org/10.1175/BAMS-D-15-00135.1
- Hourdin, F., Musat, I., Bony, S., Braconnot, P., Codron, F., Dufresne, J.-L., et al. (2006). The LMDZ4 general circulation model: Climate performance and sensitivity to parametrized physics with emphasis on tropical convection. *Climate Dynamics*, 27(7–8), 787–813. https://doi.org/10.1007/s00382-006-0158-0
- Hourdin, F., Rio, C., Grandpeix, J.-Y., Madeleine, J.-B., Cheruy, F., Rochetin, N., et al. (2020). LMDZ6A: The atmospheric component of the IPSL climate model with improved and better tuned physics. *Journal of Advances in Modeling Earth Systems*, *125*, e2019MS001892. https://doi.org/10.1029/2019MS001892
- Iacobellis, S. F., & Somerville, R. C. J. (2000). Implications of microphysics for cloud-radiation parameterizations: Lessons from TOGA COARE. Journal of the Atmospheric Sciences, 57(2), 161–183. https://doi.org/10.1175/1520-0469(2000)057<0161:IOMFCR>2.0.CO;2 Jam, A., Hourdin, F., Rio, C., & Couvreux, F. (2013). Resolved versus parametrized boundary-layer plumes. Part III: Derivation of a sta-
- tistical scheme for cumulus clouds. *Boundary-Layer Meteorology*, 147, 421–441. https://doi.org/10.1007/s10546-012-9789-3
- Jouhaud, J., Dufresne, J. L., Madeleine, J. B., Hourdin, F., Couvreux, F., Villefranque, N., & Jam, A. (2018). Accounting for vertical subgrid-scale heterogeneity in low-level cloud fraction parameterizations. *Journal of Advances in Modeling Earth Systems*, 10, 2686–2705. https://doi.org/10.1029/2018MS001379
- King, M. D., Platnick, S., Menzel, W. P., Ackerman, S. A., & Hubanks, P. A. (2013). Spatial and temporal distribution of clouds observed by MODIS onboard the Terra and Aqua satellites. *IEEE Transactions on Geoscience and Remote Sensing*, 51(7), 3826–3852. https://doi.org/ 10.1109/TGRS.2012.2227333

Klemp, J. B., & Wilhelmson, R. B. (1978). The simulation of three-dimensional convective storm dynamics. Journal of Atmospheric Sciences, 35, 1070–1096. https://doi.org/10.1175/1520-0469(1978)035<1070:TSOTDC>2.0.CO;2

Konsta, D., Dufresne, J. L., Chepfer, H., Idelkadi, A., & Cesana, G. (2016). Use of A-train satellite observations (CALIPSO-PARASOL) to evaluate tropical cloud properties in the LMDZ5 GCM. *Climate Dynamics*, 47(3-4), 1263–1284. https://doi.org/10.1007/s00382-015-2900-y Laval, K., Le Treut, H., & Sadourny, R. (1981). Effect of cumulus parameterization on the dynamics of a general circulation model.

- Geophysical and Astrophysical Fluid Dynamics, 17(1), 113–127. https://doi.org/10.1080/03091928108243676
- Le Treut, H., & Li, Z. X. (1991). Sensitivity of an atmospheric general circulation model to prescribed SST changes: Feedback effects associated with the simulation of cloud optical properties. *Climate Dynamics*, *5*, 175–187. https://doi.org/10.1007/BF00251808
- Lemonnier, F., Madeleine, J. B., Claud, C., Palerme, C., Genthon, C., L'Ecuyer, T., & Wood, N. B. (2020). CloudSat-inferred vertical structure of snowfall over the Antarctic continent. *Journal of Geophysical Research: Atmospheres*, 125, e2019JD031399. https://doi.org/ 10.1029/2019JD031399

Li, Z.-X. (1999). Ensemble atmospheric GCM simulation of climate interannual variability from 1979 to 1994. Journal of Climate, 12(4), 986–1001. https://doi.org/10.1175/1520-0442(1999)012<0986:EAGSOC>2.0.CO;2

- Liou, K. N. (2003). An introduction to atmospheric radiation. By K. N. Liou. Academic Press. Second edition, 2002. pp. xiv+583. ISBN 0 12 451451 0. Quarterly Journal of the Royal Meteorological Society, 129, 1741–1741. https://doi.org/10.1256/003590003102695746
- Loeb, N. G., Wielicki, B. A., Doelling, D. R., Smith, G. L., Keyes, D. F., Kato, S., et al. (2009). Toward optimal closure of the Earth's top-ofatmosphere radiation budget. *Journal of Climate*, 22(3), 748. https://doi.org/10.1175/2008JCLI2637.1

Lurton, T., Balkanski, Y., Bastrikov, V., Bekki, S., Bopp, L., Brockmann, P., et al. (2020). Implementation of the CMIP6 forcing data in the IPSL-CM6A-LR model. *Journal of Advances in Modeling Earth Systems*, *12*, e2019MS001940. https://doi.org/10.1029/2019MS001940

- Mauritsen, T., Stevens, B., Roeckner, E., Crueger, T., Esch, M., Giorgetta, M., et al. (2012). Tuning the climate of a global model. *Journal of Advances in Modeling Earth Systems*, *4*, M00A01. https://doi.org/10.1029/2012MS000154
- McCoy, D. T., Tan, I., Hartmann, D. L., Zelinka, M. D., & Storelvmo, T. (2016). On the relationships among cloud cover, mixed-phase partitioning, and planetary albedo in GCMs. *Journal of Advances in Modeling Earth Systems*, 8, 650–668. https://doi.org/10.1002/ 2015MS000589
- Mlawer, E. J., Taubman, S. J., Brown, P. D., Iacono, M. J., & Clough, S. A. (1997). Radiative transfer for inhomogeneous atmospheres: RRTM, a validated correlated-k model for the longwave. *Journal of Geophysical Research*, 102(D14), 16,663–16,682. https://doi.org/ 10.1029/97JD00237

Morcrette, J.-J. (1991). Radiation and cloud radiative properties in the European Centre for Medium Range Weather Forecasts forecasting system. *Journal of Geophysical Research*, 96(D5), 9121–9132. https://doi.org/10.1029/89JD01597

Morcrette, J.-J., & Fouquart, Y. (1986). The overlapping of cloud layers in shortwave radiation parameterizations. *Journal of Atmospheric Sciences*, 43(4), 321–328. https://doi.org/10.1175/1520-0469(1986)043<0321:TOOCLI>2.0.CO;2

Nam, C., Bony, S., Dufresne, J. L., & Chepfer, H. (2012). The 'too few, too bright' tropical low-cloud problem in CMIP5 models. *Geophysical Research Letters*, 39, L21801. https://doi.org/10.1029/2012GL053421

- O'Dell, C. W., Wentz, F. J., & Bennartz, R. (2008). Cloud liquid water path from satellite-based passive microwave observations: A new climatology over the global oceans. *Journal of Climate*, 21(8), 1721. https://doi.org/10.1175/2007JCLI1958.1
- Rio, C., Del Genio, A. D., & Hourdin, F. (2019). Ongoing breakthroughs in convective parameterization. *Current Climate Change Reports*, 5(2), 95–111. https://doi.org/10.1007/s40641-019-00127-w
- Rio, C., & Hourdin, F. (2008). A thermal plume model for the convective boundary layer: Representation of cumulus clouds. Journal of Atmospheric Sciences, 65(2), 407–425. https://doi.org/10.1175/2007JAS2256.1
- Rochetin, N., Grandpeix, J.-Y., Rio, C., & Couvreux, F. (2014). Deep convection triggering by boundary layer thermals. Part II: Stochastic triggering parameterization for the LMDZ GCM. *Journal of Atmospheric Sciences*, 71(2), 515–538. https://doi.org/10.1175/JAS-D-12-0337.1

Sadourny, R. (1975). Compressible model flows on the sphere. *Journal of Atmospheric Sciences*, 32(11), 2103–2110. https://doi.org/10.1175/1520-0469(1975)032<2103:CMFOTS>2.0.CO;2

- Sadourny, R., & Laval, K. (1984). January and July performance of the LMD GCM. New perspectives in climate modelling (developments in atmospheric science 16). *Journal of Climatology*, 4(5), 566–566. https://doi.org/10.1002/joc.3370040513
- Schlesinger, M. E., Oh, J.-H., & Rosenfeld, D. (1988). A parameterization of the evaporation of rainfall. Monthly Weather Review, 116(10), 1887. https://doi.org/10.1175/1520-0493(1988)116<1887:APOTEO>2.0.CO;2
- Smith, E. A., & Shi, L. (1992). Surface forcing of the infrared cooling profile over the Tibetan Plateau. Part I: Influence of relative longwave radiative heating at high altitude. *Journal of Atmospheric Sciences*, 49(10), 805–822. https://doi.org/10.1175/1520-0469(1992)049<0805: SFOTIC>2.0.CO;2



- Sulak, A., Calabrase, W., Ryan, S. D., & Heus, T. (2020). The contributions of shear and turbulence to cloud overlap for cumulus clouds. Journal of Geophysical Research: Atmospheres, 125, e2019JD032017. https://doi.org/10.1029/2019JD032017
- Teixeira, J., Cardoso, S., Bonazzola, M., Cole, J., Delgenio, A., Demott, C., et al. (2011). Tropical and subtropical cloud transitions in weather and climate prediction models: The GCSS/WGNE Pacific Cross-Section Intercomparison (GPCI). Journal of Climate, 24(20), 5223–5256. https://doi.org/10.1175/2011JCLI3672.1
- Vial, J., Dufresne, J.-L., & Bony, S. (2013). On the interpretation of inter-model spread in CMIP5 climate sensitivity estimates. Climate Dynamics, 41(11-12), 3339–3362. https://doi.org/10.1007/s00382-013-1725-9
- Vignon, E., Hourdin, F., Genthon, C., Gallée, H., Bazile, E., Lefebvre, M.-P., et al. (2017). Antarctic boundary layer parametrization in a general circulation model: 1-D simulations facing summer observations at Dome C. Journal of Geophysical Research: Atmospheres, 122, 6818–6843. https://doi.org/10.1002/2017JD026802
- Webb, M., Senior, C., Bony, S., & Morcrette, J. J. (2001). Combining ERBE and ISCCP data to assess clouds in the Hadley Centre, ECMWF and LMD atmospheric climate models. *Climate Dynamics*, 17(12), 905–922. https://doi.org/10.1007/s003820100157
- Xie, S., Lin, W., Rasch, P. J., Ma, P.-L., Neale, R., Larson, V. E., et al. (2018). Understanding cloud and convective characteristics in Version 1 of the E3SM atmosphere model. *Journal of Advances in Modeling Earth Systems*, *10*, 2618–2644. https://doi.org/10.1029/2018MS001350
- Yamada, T. (1983). Simulations of nocturnal drainage flows by a q^2l turbulence closure model. *Journal of Atmospheric Sciences*, 40(1), 91–106. https://doi.org/10.1175/1520-0469(1983)040<0091:SONDFB>2.0.CO;2